Author’s response to reviews

Title: Prevalence and Causes of Blindness, Visual Impairment Among Different Ethnical Minority Groups In Xinjiang Uygur Autonomous Region, China

Authors:
Yanping Li (5861623@qq.com)
Wenyong Huang (1220792380@qq.com)
Qoyun Qiqige (aoyunqqg@163.com)
Hongwei Zhang (zhwzhw.123@163.com)
Ling Jin (lingjin41@yahoo.com)
Pula Ti (anrabc@126.com)
Jennifer Yip (jennifer.yip@lshtm.ac.uk)
Baixiang Xiao (xiaobaixiang2006@126.com)

Version: 3 Date: 24 Jan 2018

Author’s response to reviews:

1. In the Declarations, please include statements on ethics approval and consent in the 'Ethics approval and consent to participate'.

   This is done at the declarations.

2. We note that you obtained verbal consent from the participants in your study. Please state the reason why verbal consent was obtained and whether the ethics committee approved this procedure. Please include this statement in the 'Ethics approval and consent to participate' section of the Declarations.

   It’s done

3. We note that the "Data of this study is available at the RAAB Repository". Please also include a web link to the data used in this study in the 'Data availability' section of the Declarations.

   This is now added as www.raabdata.info/repository
4. We note that 'AY' has been included in the Authors' Contributions, however, 'AY' does not match any of the authors listed on this manuscript. Please clarify whether this is a typing error, or if 'AY' needs to be added to the author list.

That should be the same as Aoyun Qiqige and abbreviation should be AQ, sorry for the typo.

If you need to add 'AY' to the author list, you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

5. We note that 'figure 1' is cited in the manuscript, but, unfortunately, seems to be missing from the manuscript. Please ensure to upload the figure as a separate file with your revised manuscript. Please also ensure that you provide a figure legend for figure 1 under a separate heading of 'Figure Legends' after the References.

Figure 1 is attached and the Figure legends is now added after references.

6. Please remove 'Comments for revision 10Jan2018-Xiao responded.docx' from the file inventory, as this is no longer needed.

It is now deleted.

7. We note that some of your references are not formatted correctly. Please ensure that these are re-formatted according to the BioMed Central reference style.

They are now reformatted.

All web links and URLs should be provided in full, including both the title of the site and the URL, as well as the date the site was accessed, in the following format: The Mouse Tumor Biology Database. http://tumor.informatics.jax.org/mtbwi/index.do. Accessed 20 May 2013. If an author or group of authors can clearly be associated with a web link (e.g. for blogs) they should be included in the reference.


for more guidance on referencing, please visit: https://bmcophthalmol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article
Please also take a moment to check our website at https://boph.editorialmanager.com/ for any additional comments that were saved as attachments.

Once you have made the necessary corrections, please submit a revised manuscript online at:

https://boph.editorialmanager.com/

If you have forgotten your password, please use the 'Send Login Details' link on the login page at https://boph.editorialmanager.com/. For security reasons, your password will be reset.

We request that a point-by-point response letter accompanies your revised manuscript. This letter must provide a detailed response to each reviewer/editorial point raised, describing what amendments have been made to the manuscript text and where these can be found (e.g. Methods section, line 12, page 5). If you disagree with any comments raised, please provide a detailed rebuttal to help explain and justify your decision.

Please also ensure that your revised manuscript conforms to the journal style, which can be found at the Submission Guidelines on the journal homepage.

A decision will be made once we have received your revised manuscript, which we expect by 26 Jan 2018.

Please note that you will not be able to add, remove, or change the order of authors once the editor has accepted your manuscript for publication. Any proposed changes to the authorship must be requested during peer-review, and adhere to our criteria for authorship as outlined in BioMed Central's policies. To request a change in authorship, please download the 'Request for change in authorship form' which can be found here - http://www.biomedcentral.com/about/editorialpolicies#authorship. Please note that incomplete forms will be rejected. Your request will be taken into consideration by the editor, and you will be advised whether any changes will be permitted. Please be aware that we may investigate, or ask your institute to investigate, any unauthorized attempts to change authorship or discrepancies in authorship between the submitted and revised versions of your manuscript.

We look forward to receiving your revised manuscript and please do not hesitate to contact us if you have any questions.

Best wishes,

Amy Branch-Hollis
BMC Ophthalmology
https://bmcophthalmol.biomedcentral.com/
BMC Ophthalmology operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Reviewer reports:

Shin Goo Park (Reviewer 1): no comments Thanks

Wool Suh (Reviewer 2): Authors revised the manuscript properly.

There is no more recommendation on the context of manuscript.

However, a few English grammatical errors are noticed, authors needs to correct some errors before publication. Rechecked and revised. Thanks